Author's response to reviews

Title: A proof of principle for using adaptive testing in Routine Outcome Monitoring: the efficiency of the Mood and Anxiety Symptoms Questionnaire - Anhedonic Depression CAT

Authors:

Niels Smits (n.smits@psy.vu.nl)
Frans G Zitman (f.g.zitman@lumc.nl)
Pim Cuijpers (p.cuijpers@vu.nl)
Margien E den Hollander (m.e.den_hollander@lumc.nl)
Ingrid VE Calier (i.v.e.carlier@lumc.nl)

Version: 4 Date: 10 January 2012

Author's response to reviews:

Dear Sir/Madam,

As requested by you, I changed the following:

Figures: I cited the figures using the citation number whenever I did not use 'Figure 1' or 'Figure 2' as [figure 1] and [figure 2].

Yours Sincerely,

N Smits